Author's response to reviews

Title: The influence of academic examinations on energy and nutrient intake in male university students

Authors:

Margo E Barker (m.e.barker@sheffield.ac.uk)
Richard J Blain (richard.blain2@dcu.ie)
Jean M Russell (j.russell@sheffield.ac.uk)

Version: 2 Date: 11 March 2015

Author's response to reviews:

The following sections have been added: title page, keywords, conclusion, availability of supporting data, list of abbreviations, competing interests, authors' contribution, acknowledgements and tables. These have been done using track changes